Author’s response to reviews

Title: Characterization of gut microbiota in children with pulmonary tuberculosis

Authors:

Weiran Li (lwr031@126.com)
Yu Zhu (zhuyu_wj@163.com)
Qiong Liao (33765410@qq.com)
Zhiling Wang (wang_zhi_ling@126.com)
Chaomin Wan (wcm0220@126.com)

Version: 4 Date: 09 Oct 2019

Author’s response to reviews:

thanks for your attention and we have revised our manuscript according to your comments

1. Written informed consent was obtained from guardians of participants prior to study enrollment. we have added this content on the Ethics approval and consent to participate part

2. we have represented authors' names using their full initials and included a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript

3. we removed the reply letter

4. the manuscript was single, final, clean version that we have submitted.

5. the name of our hospital is west china second university hospital, Sichuan university. in our paper, we have used this standard name.